Author's response to reviews

Title: Systematic review of worldwide variations of the prevalence of wheezing symptoms in children

Authors:

Swatee P Patel (s.p.patel@gre.ac.uk)
Marjo-Riitta Jarvelin (m.jarvelin@imperial.ac.uk)
Mark P Little (mark.little@imperial.ac.uk)

Version: 6 Date: 10 November 2008

Author's response to reviews:

The changes requested to the references and to the figure have all been made.